Author's response to reviews

Title: Assessing perceived well-being among elderly people

Authors:

Jori S Reijula (jori.reijula@tkk.fi)
Toni Rosendahl (toni.rosendahl@gmail.com)
Kari E Reijula (kari.reijula@ttl.fi)
Paula Roilas (paula.roilas@terveyspalvelu.fi)
Heikki Roilas (heikki.roilas@lappeenranta.fi)
Raimo Sepponen (raimo.sepponen@tkk.fi)

Version: 2 Date: 21 July 2009

Author's response to reviews:

Dear Editor,

I have made the changes you requested, explained below:

(1) I have changed reference number 10 to reflect that my article has been published.
(2) I removed the indication of the equal contribution of the 5 authors, since all 5 authors didn't contribute exactly equally to my article.
(3) I changed the "Competing interests" section so that it now includes a remark of the pending patent application for "Con-Dis" device.
(4) The ethics committee requested is now mentioned in the "methods" section.

I hope these revisions are satisfactory.

Kindly,

Jori Reijula, MSciTechn